Author's response to reviews

Title: In vitro antioxidant and anticancer effects of solvent fractions from Prunella vulgaris var. lilacina

Authors:

Yu-Jin Hwang (yujinh21@skku.edu)
Eun-Ju Lee (bono2180@naver.com)
Haeng-Ran Kim (kimhr@korea.kr)
Kyung-A Hwang (kah366@korea.kr)

Version: 4 Date: 24 October 2013

Author's response to reviews: see over
October 24, 2013

Dear James Prozenko,

Thank you very much for your effort.

Enclosed please find our revised figures and text of the manuscript (ID 6807539991032192) with the title “In vitro antioxidant and anticancer effects of solvent fractions from Prunella vulgaris var. lilacina.”

Below you will find our point-by-point responses to the referee’s comments. On behalf of the other authors, I hope that our revised manuscript will be judged acceptable for publication in BMC Complementary and Alternative Medicine.

Sincerely,

Kyung-A Hwang, Ph. D.

Kyung-A Hwang, Ph. D.
National Academy of Agricultural Science, RDA
160, Nokjiro, Gwonseon-gu, Suwon, Gyeonggi-do 441-853, Republic of Korea
Tel: +82-31-299-0527, Fax: +82-31-299-0504, E-mail: kah366@korea.kr
This study is interesting and helpful for the advance in the antioxidant and anticancer activities of this traditional plant medicine. The manuscript is suitable for publication. Please revise according to review comment.

Review comment:

About the plant latin name “Prunella vulgaris var. lilacina”, the word “var.” do not need use italics. Please check again.

The paper has been conditionally accepted.

RESPONSE: We re-checked the plant latin name in manuscript and corrected “Prunella vulgaris var. lilacina” instead of “Prunella vulgaris var. lilacina”.